Author's response to reviews

**Title:** Weight change during chemotherapy changes the prognosis in non metastatic breast cancer for the worse

**Authors:**

Emilie Thivat (Emilie.Thivat@cip.fr)
Sophie Thérondel (therondelsophie@yahoo.fr)
Olivier Lapirot (lapirot_olivier@yahoo.fr)
Catherine Abrial (Catherine.ABRIAL@cip.fr)
Pierre Gimbergues (Recherche.CLINIQUE@cip.fr)
Emilie Gadéa (emilie.gadea@hotmail.fr)
Eloïse Planchat (Eloise.Planchat@cip.fr)
Fabrice Kwiatkowski (Fabrice.Kwiatkowski@cip.fr)
Marie A Mouret-Reynier (Marie-Ange.Mouret-Reynier@cip.fr)
Philippe Chollet (Philippe.Chollet@cip.fr)
Xavier Durando (xavier.DURANDO@cip.fr)

**Version: 5 Date: 12 November 2010**

**Author's response to reviews:**

Dear Sir,

Please find enclosed a revised manuscript MS 9851151553337963 entitled: “Weight change during chemotherapy is a poor prognostic in non metastatic breast cancer”, that would like to submit for publication in BMC Cancer.

The manuscript is a resubmission following revisions. formatting changes have been performed according to the reviewer recommendations: the list of the author in the manuscript is written exactly as they are in the submission system, the sentence "all authors read and approved the final manuscript" has been added in the authors contributions section. Moreover, the visible vertical lines from the tables have been removed, the figures are cropped as closely possible.

We thank you very much for considering this manuscript and remain,

Yours sincerely.

Emilie THIVAT.